Author's response to reviews

Title: Longitudinal study on visual outcome and spectacle use after intracapsular cataract extraction in Northern India

Authors:

Sanjeev Gupta (comoph@ndfvsnl.net.in)
Gvs Murthy (comoph@ndfvsnl.net.in)
Namrata Sharma (namrata@hotmail.com)

Version: 4 Date: 2 Jun 2003

Dear Editor
Thanks a lot for accepting our manuscript. The following changes in the manuscript relating to the formatting have been made:
- The Qualifications of the authors have been removed from the title page.
- The Background section has been expanded to discuss the context and purpose of the study
- Manuscript sections includes the following sections as suggested: Abstract, Background, Methods, Results, Discussion, Conclusions, Competing Interests, Authors' Contributions, Acknowledgements, References, Tables
- The highlighted text has been made into normal type to be consistent with the rest of the manuscript.
- The references have been cited in the text using consecutive numbers in square brackets.
- Competing Interests (medicine are none and have been declared.

Thanking You
Yours sincerely
Namrata Sharma